Author's response to reviews

Title: Synoptic Tool for reporting of Hematological and Lymphoid neoplasms based on World Health Organization classification and College of American Pathologists checklist

Authors:

Sambit K Mohanty (mohantys2@upmc.edu)
Anthony L Piccoli (piccolial@upmc.edu)
Lisa J Devine (devinelj@upmc.edu)
Ashokkumar A Patel (patelaas@upmc.edu)
William C Gross (grosswc@upmc.edu)
Steven H Swerdlow (swerdlows@upmc.edu)
Sharon B Winters (wintersssb@upmc.edu)
Michael J Becich (becich@pitt.edu)
Anil V Parwani (parwaniav@upmc.edu)

Version: 2 Date: 10 January 2007

Author's response to reviews:

Name of one of the authors was incorrect.